Author's response to reviews

Title: Concurrence of villous adenoma and non-muscle invasive bladder cancer arising in the bladder: a case report and review of the literature

Authors:

Yoichiro Kato (j2c789@bma.biglobe.ne.jp)
Susumu Konari (konyari99@yahoo.co.jp)
Wataru Obara (watao@iwate-med.ac.jp)
Tamotsu Sugai (tsugai@iwate-med.ac.jp)
Tomoaki Fujioka (tomof@iwate-med.ac.jp)

Version: 5 Date: 25 June 2013

Author's response to reviews: see over
June 25, 2013

Dr. Hayley Henderson

Executive Editor
BMC Urology
Editorial Office
Floor 6, 236 Gray's Inn Road
London WC1X 8HB United Kingdom


Dear Dr. Henderson,

Thank you for reviewing and accepting our article for submission titled “Concurrence of villous adenoma and non-muscle invasive bladder cancer arising in the bladder: a case report and review of the literature”.

I received notification of acceptance on June 18, 2013.

Please find the first revised and the latest revised manuscript.
As you notice, we have revised the manuscript by modifying the Author’s contributions and Acknowledgements sections, based on the comments made by the BMC Urology Editorial Team.

Please also find our point-by-point response including below to the comments raised by the BMC Urology Editorial Team. Accordingly, we attached the first revised manuscript marked with all the changes made during the revision process indicated by shading. The new text is underlined and in a red color in the response.
As you notice, we agreed and complied with all the comments. Moreover, we submitted the final manuscript.

Please contact me if you have any questions about the article or the materials I have enclosed. Thank you again for your time and this opportunity.

Sincerely Yours,
Yoichiro Kato,
Department of Urology, Iwate Medical University,
Uchimaru, Moriokashi, Iwate, 020-8505, Japan.
Phone: 019-651-5111 (8489)
Fax: 019-623-1527
Email: j2c789@bma.biglobe.ne.jp
Comments from the BMC Urology Editorial Team

**Comment 1.** 1. Please include an Acknowledgements section in your manuscript

**Response 1.** Thank you for your comment. We revised from “TS performed histopathological examinations. Language editor Paul Langman edited our manuscript. All authors…” to “TS performed histopathological examinations. Linguage editor Paul Langman edited our manuscript. All authors…” on page12. Moreover, we made an Acknowledgements section as follows, “Acknowledgements Language editor Paul Langman edited our manuscript.” on page12.